Author's response to reviews

Title: Work-family conflict and burnout among Chinese female nurses: the mediating effect of psychological capital

Authors:

Yang Wang (allylove666@yahoo.com.cn)
Ying Chang (queency99@126.com)
Jialiang Fu (fujialiang@126.com)
Lie Wang (liewang@mail.cmu.edu.cn)

Version: 6 Date: 28 September 2012

Author's response to reviews:

Dear Editor:

This is the final revised version of manuscript "Work-family conflict and burnout among Chinese female nurses: The mediating effect of psychological capital". Revisions were made based on the suggestions from Publication Team.

Thank you for you to consider it for publication as a “Research Article” in “BMC Public Health”.

Best wishes,
Sincerely yours,
Lie Wang
China Medical University